Author's response to reviews

Title: Pain and satisfaction during rigid cystoscopic ureteral stent removal: A preliminary study

Authors:

Jae Heon Kim (piacekjh@hanmail.net)
Sun Young Park (sunnypark97@schmc.ac.kr)
Moon Gyu Kim (rolansrb1@schmc.ac.kr)
Hoon Choi (doc71377@naver.com)
Dan Song (backsa7@schmc.ac.kr)
Seong Woo Cho (orsay2000@mac.com)
Yun Seob Song (yssong@schmc.ac.kr)

Version: 5 Date: 4 July 2014

Author's response to reviews: see over
1. Clinical Trial
Please note that BioMed Central requests a trial registration number for manuscripts reporting work that falls within the International Committee of Medical Journal Editors (ICMJE) definition of a clinical trial: any research study that prospectively assigns human subjects to one or more health related interventions to evaluate the effects on health outcomes. Your study describes three different pain relief interventions during ureteral stent removal, and the health outcomes in each case. As such, your study qualifies as a clinical trial. Before we can accept your manuscript for publication, we would like you to confirm that your clinical trial is in a publicly accessible registry. We only accept registration numbers issued by registries that meet all of the ICMJE criteria (http://www.icmje.org/publishing_10register.html). Registries which meet the requirements of the ICMJE include WHO Primary Registries (http://www.who.int/ictrp/network/primary/en/index.html). If you have not already registered your trial, you are able to do so retrospectively. The trial registration number should be included as the last line of the abstract of the manuscript.

-> We added the registry number of IRB at the bottom of abstract.

2. Consent to Participate
Informed consent must be documented in all studies involving human participants. Please can you confirm in the Methods section whether patient consents were obtained.

-> Thank you for your great comment. According to your suggestion, we have added the document about patient consent.

3. CONSORT
To ensure that your reporting is transparent and reproducible, we would appreciate if you downloaded the CONSORT checklist (http://www.consort-statement.org/) and revise your manuscript accordingly.

-> Thank you for your great comment. This study is an interventional study which could not be done by total blinding or random allocation. About the sample size, we hypothesized that 12 patients for each group would be sufficient for analysis (sample size of 12 per group rule of thumb for a pilot study: pharmaceutical statistics. 2005;4:287-291)

4. Formatting
Please can you go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points. The link to the formatting checklist is provided at http://www.biomedcentral.com/info/ifora/medicine_journals

-> Thank you for your kind comment. We have checked your formatting checklist again.

5. Questionnaire
We assume that your original questionnaire was in Korean and has been translated for the purposes of this study. If this is true, we would ask you to revise the phrases "much satisfied" to "very satisfied" and "so so" to "average".

-> Thank you for your kind comments. According to your suggestion, we have
changed those words.

6. Copyedit
You now have the opportunity to copyedit your manuscript to improve the quality of English. Please bear in mind that as we are a free-access publisher, we cannot bear the costs of copyediting English ourselves. Please ensure particular attention is paid to the abstract. In particular, the recently added sentence is not grammatically correct. We have revised this for you, below:

-> Thank you for your kind comment. Except the changed sentences in abstract, whole text was edited by English editing service.